Author’s response to reviews

Title: Impact of diabetes on sarcopenia and mortality in patients undergoing hemodialysis

Authors:
Katsuhito Mori (ktmori@med.osaka-cu.ac.jp)
Kozo Nishide (westoutdedede@yahoo.co.jp)
Senji Okuno (okuno@shirasagi-hp.or.jp)
Tetsuo Shoji (t-shoji@med.osaka-cu.ac.jp)
Masanori Emoto (memoto@med.osaka-cu.ac.jp)
Akihiro Tsuda (naranotsudadesu@infoseek.jp)
Shinya Nakatani (m2026719@med.osaka-cu.ac.jp)
Yasuo Imanishi (imanishi@med.osaka-cu.ac.jp)
Eiji Ishimura (ish@med.osaka-cu.ac.jp)
Tomoyuki Yamakawa (yamakawa@shirasagi-hp.or.jp)
Shigeichi Shoji (shoji@shirasagi-hp.or.jp)
Masaaki Inaba (m1356849@med.osaka-cu.ac.jp)

Version: 4 Date: 26 Feb 2019

Author’s response to reviews:

Dear Prof. Jing Zhang,

Thank you very much for assessing our manuscript. According to your suggestion, we have revised the manuscript. I hope you now find that this version is suitable for publication.

Thank you very much for your help.
Sincerely yours,

Katsuhito Mori